Author's response to reviews

Title: Shear wave elastography in chronic kidney disease: a pilot experience in native kidneys

Authors:

Anthony E Samir (asamir@mgh.harvard.edu)
Andrew S Allegretti (aallegretti@partners.org)
Qingli Zhu (qzhu@mgh.harvard.edu)
Manish Dhyani (Dhyani.Manish@mgh.harvard.edu)
Arash Anvari (AANVARI@PARTNERS.ORG)
Dorothy A Sullivan (DSULLIVAN18@mgh.harvard.edu)
Caitlin A Trottier (caittrot@bu.edu)
Sarah Dougherty (SDOUGHERTY@MGH.HARVARD.EDU)
Winfred W Williams (WWWILLIAMS@mgh.harvard.edu)
Jodie L Babitt (Babitt.Jodie@mgh.harvard.edu)
Julia Wenger (JWENCER1@partners.org)
Ravi I Thadhani (RTHADHANI@mgh.harvard.edu)
Herbert Y Lin (Lin.Herbert@mgh.harvard.edu)

Version: 3 Date: 29 May 2015

Author's response to reviews:

The following changes (requested May 29, 2015 by the Editors) have been made:

Editorial Request:
1. Please include the name of the ethics committee in the Methods section.
2. Please clarify if consent was written or oral.
3. Please include funding information in the Acknowledgements and include any consultancy fees and equipment donations in the Conflict of Interest
4. Move tables and figure legends to after the References